Author's response to reviews

Title: Prevalence and socio-demographic correlates of stunting and thinness among Pakistani primary school children

Authors:

Muhammad Umair Mushtaq (mushtaqmu@gmail.com)
Sibgha Gull (sibgha_gul@ymail.com)
Usman Khurshid (usmankhurshid_aimc@yahoo.com)
Ubeera Shahid (mushtaqmu@gmail.com)
Mushtaq Ahmad Shad (mushtaq_shad@yahoo.com)
Arif Mahmood Siddiqui (rdsidiqi@gmail.com)

Version: 4 Date: 5 October 2011

Author's response to reviews: see over
Honorable
Editor-in-Chief
BMC Public Health

We are pleased to submit the final version of our manuscript numbered MS 1078053693570407 titled “Prevalence and socio-demographic correlates of stunting and thinness among Pakistani primary school children.”

The requested formatting changes have been made as indicated below:

...

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Figures: All the figures are listed as suggested.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: The figures are prepared as per online figure guide

Typography: Manuscript is checked for any typographical and content errors.

...

It has been a great experience to publish with you and we highly appreciate the time and efforts of the editorial team.

Regards

Dr Muhammad Umair Mushtaq, MBBS, MACP
(Principal Investigator & Corresponding Author)